

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Eshun, Samuel Fiifi; Kočenda, Evžen

Working Paper Money talks, green walks: Does financial inclusion promote green sustainability in Africa?

IES Working Paper, No. 23/2024

Provided in Cooperation with: Charles University, Institute of Economic Studies (IES)

Suggested Citation: Eshun, Samuel Fiifi; Kočenda, Evžen (2024) : Money talks, green walks: Does financial inclusion promote green sustainability in Africa?, IES Working Paper, No. 23/2024, Charles University in Prague, Institute of Economic Studies (IES), Prague

This Version is available at: https://hdl.handle.net/10419/301126

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



MONEY TALKS, GREEN WALKS: DOES FINANCIAL INCLUSION PROMOTE GREEN SUSTAINABILITY IN AFRICA?

p)

Samuel Fiifi Eshun Evžen Kočenda

IES Working Paper 23/2024

Institute of Economic Studies,
Faculty of Social Sciences,
Charles University in Prague
[UK FSV – IES]
Opletalova 26
CZ-110 00, Prague
E-mail : ies@fsv.cuni.cz
http://ies.fsv.cuni.cz
Institut ekonomických studií
Fakulta sociálních věd
Univerzita Karlova v Praze
Opletalova 26
110 00 Praha 1

E-mail : ies@fsv.cuni.cz http://ies.fsv.cuni.cz

Disclaimer: The IES Working Papers is an online paper series for works by the faculty and students of the Institute of Economic Studies, Faculty of Social Sciences, Charles University in Prague, Czech Republic. The papers are peer reviewed. The views expressed in documents served by this site do not reflect the views of the IES or any other Charles University Department. They are the sole property of the respective authors. Additional info at: <u>ies@fsv.cuni.cz</u>

Copyright Notice: Although all documents published by the IES are provided without charge, they are licensed for personal, academic or educational use. All rights are reserved by the authors.

Citations: All references to documents served by this site must be appropriately cited.

Bibliographic information:

Eshun S.F., Kočenda E. (2024): "Money Talks, Green Walks: Does Financial Inclusion Promote Green Sustainability in Africa?" IES Working Papers 23/2024. IES FSV. Charles University.

This paper can be downloaded at: http://ies.fsv.cuni.cz

Money Talks, Green Walks: Does Financial Inclusion Promote Green Sustainability in Africa?

Samuel Fiifi Eshun¹ Evžen Kočenda²

¹Institute of Economic Studies, Charles University, Prague, Czech Republic Corresponding author: 26453053@fsv.cuni.cz ² Institute of Economic Studies, Charles University, Prague, Czech Republic; CESifo, Munich, Germany; IOS, Regensburg, Germany; Institute of Information Theory and Automation of the CAS, Prague; the Euro Area Business Cycle Network

May 2024

Abstract:

This study explores the dynamic relationship between financial inclusion and green sustainability across 38 African countries. We constructed an environmental pollution index and a financial inclusion index covering the period 2000-2021 to account for the several dimensions within both indicators and employed them in the System GMM approach. We also tested for intra-regional heterogeneity in Africa. Our empirical results show that financial inclusion, while economically beneficial, poses a significant risk of environmental degradation and has a distinctive inverted U-shaped relationship. A direct link between increases in financial inclusion and pollution alters at a turning point, beyond which further increments in financial inclusion enhance green sustainability. The same pattern is observed for aggregate output. The results hold even when we control for a score of macro-level determinants. Our findings indicate the existence of an intra-regional heterogeneity in that Southern and Western African states exhibit a more significant negative impact on environmental pollution than Eastern Africa. These results remain robust for alternative proxies of green sustainability. We offer valuable insights for policymakers to promote sustainability through inclusive financial practices and policies in Sub-Saharan Africa.

JEL: C23, E44, F64, K32, O55, Q43

Keywords: Environmental Pollution Index, Financial Inclusion Index, Green Sustainability, Sub-Saharan Africa (SSA), System Generalized Methods of Moments (GMM) **Acknowledgement:** We are grateful for the useful comments from Alessandro Chiari, Christian Vossler, Emmanuel Amoah, Harriet Kwartemaa, Isaac Nunoo, Milan Ščasný, Tersoo Lorgurum, and presentation participants in Prague. This work was supported by the Grantova Agentura Ceske Republiky under Grant No 19-26812X.

1. Introduction, motivation, and related literature

In recent years, climate change, global warming and other threats to the environment have forced the world, including Africa, to look deeper into sustainability (Udeagha & Muchapondwa, 2022). The African continent, endowed with rich natural resources and diverse ecosystems, faces this dual challenge of fostering economic growth while safeguarding its environmental heritage (Opoku, Acheampong et al., 2022). This challenge includes high levels of carbon emissions, enormous environmental pollution, and an increasing dependence on energy sources. The exploitation of its natural resources (Hsu et al., 2023) and the expansion of economic activities (Ozturk & Ullah, 2022) raises significant environmental challenges. As such, the intersection of economic development and environmental degradation presents a critical area of study in financial inclusion. Grounded in the Resource Mobilisation Theory (McCarthy & Zald, 1977), financial inclusion is pivotal in mobilising funds from a broader population and channelling those funds into productive investments. Hence, since fund accessibility influences income and consumption patterns of individuals or organisations (Ait Lahcen & Gomis-Porqueras, 2021), we ask whether it contributes to sustainable development or worsening environmental challenges. In this paper, we seek to investigate the impact of financial inclusion on green sustainability in Africa, where green sustainability reflects the extent of environmental pollution. By examining this relationship, we aim to provide insights into how financial inclusion can be leveraged to promote sustainable development across the continent.

The term "green sustainability" denotes a concept that primarily emphasises environmental consciousness, practices, and adopting eco-friendly products and services to minimise negative impacts on ecosystems and safeguard the environment for future generations. It is often synonymous with the idea of environmental sustainability (Udeagha & Muchapondwa, 2023). Environmental sustainability encompasses a range of practices, including sustainable transportation, investing in waste-to-energy projects, green building designs, reducing greenhouse gas emissions, conserving biodiversity, promoting renewable energy, and adopting sustainable production and consumption patterns. However, the extent to which financial inclusion can influence this pursuit remains a subject of empirical inquiry. Financial inclusion, in essence, provides affordable and accessible financial products and services to a broader cross-section of society, including individuals traditionally excluded from the formal financial system. The increase in financial accessibility and usage promotes reduced transaction costs, the development

of innovative financial products, and the increased coverage of financial services (Beck, 2020). Drawing from the Resource Mobilisation Theory, an increase in financial accessibility is characterised by urbanisation, industrialisation, and economic affluence.

The theory suggests that by expanding access to financial services, a broader base of individuals can participate in economic activities, thereby contributing to the mobilisation of financial resources. Hence, the influence of financial inclusion on environmental pollution can be understood through its impact on funding channels. It empowers (1) companies with access to credit to import heavy machinery for new projects (Ozturk & Ullah, 2022), (2) individuals to finance luxury and energy-intensive or energy-saving equipment and appliances (Ozturk & Acaravci, 2013), and (3) an overall increase in the consumption and production of goods and services (Hafeez et al., 2019). Each of these activities, in its way, produces an impact on the environment. As a result, there is a potential for a diversified and expanded pool of funds that can be directed toward environmentally sustainable initiatives and investments.

According to Le et al. (2020), the impact of financial inclusion on the environment could either take a positive or negative trajectory. The positive facet is evident in how financial inclusion provides access to funds for investments in green technologies, fostering sustainable agricultural practices, clean-energy innovations, and energy-efficient infrastructure like solar panels and wind turbines (Usman et al., 2021). These endeavours reduce reliance on fossil fuels, mitigate greenhouse gas emissions, and strengthen overall environmental sustainability (Ji, Chen, et al., 2021; Ji, Umar, et al., 2021). On the other hand, Le et al. (2020), Frankel and Romer (1999), and Jensen (1996) argue that while access to financial services stimulates manufacturing, industrialisation, and economic activities, its consequences encompass increased carbon dioxide (CO2) emissions, heightened consumption of energy-intensive consumer goods, and augmented demand for polluting energy sources. Considering these arguments, the imperative of balancing financial inclusion with sustainability remains a critical concern, particularly in the African context.

Previous research has explored how financial inclusion or financial development affects carbon emissions in countries like China, Asia, BRICS, and Organisation for Economic Cooperation and Development (OECD) countries (Wang et al., 2022; Liu et al., 2022; Qin et al., 2021; Zaidi et al., 2021; Le et al., 2020). Wang et al. (2022) indicated in their study in China that the development of financial inclusion has a non-linear effect on environmental quality and sustainability, implying that only after reaching a certain degree of development can financial inclusion improve sustainability outcomes as such financial inclusion observes the Environmental Kuznets Curve (EKC) hypothesis in China. However, in the broader Asian region, Hussain et al. (2021) and Le et al. (2020) suggest a potential negative impact on sustainability. Ahmad et al. (2022) revealed that financial inclusion contributes to environmental pollution in the BRICS countries. Dou and Li (2022) extend this understanding by establishing a positive impact of financial inclusion on emissions. However, in the context of energy, where green sustainability is accounted for as an element of energy efficiency. Dai et al. (2022) show that financial inclusion positively impacts renewable energy efficiency for Regional Comprehensive Economic Partnership (RCEP) economies. Notably, digital financial inclusion also supports green sustainability and development by driving technological innovation and the advancement of industrial structures (Wang et al., 2022). According to Lessmann and Kalkuhl (2023), costly financial intermediation creates high-interest rate spreads, which reduces carbon emissions due to lower overall economic growth and increased abatement costs.

The growing research in these areas suggests that financial inclusion can positively and negatively affect the environment, depending on the specific context and factors such as technological innovation and green openness. However, there is a significant gap in understanding this relationship within the specific context of Africa. To the best of our knowledge, no study has delved into the role of financial inclusion in environmental pollution within the unique context of Africa, specifically Sub-Saharan Africa (SSA). Moreover, when it comes to environmental sustainability, few studies have considered its multidimensionality, such as carbon footprint, ecological footprint, and air quality, and how these factors collectively contribute to overall sustainability. The multidimensional perspective is crucial for a comprehensive understanding of sustainability. To assess the impact of financial inclusion, we test the following null hypothesis: *Financial inclusion does not influence environmental pollution levels in SSA*.

The issue under research is quite relevant, considering that SSA is increasingly home to some of the world's fastest-growing economies and large organisations (Melo & Solleder, 2022). The region is poised to suffer the most from further global warming due to its position as one of the hottest regions on earth (Asafu-Adjaye, 2014). Figure 1 underlines the considerable variations in emission intensity levels among African nations and the startling fact that carbon emissions from energy sources recorded an average annual growth rate of 1.4% from 2012 to 2022 (Energy

Institute, 2023)¹. Furthermore, financial inclusion is expanding with regional integration and globalisation (Demirgüç-Kunt et al., 2022). As of 2021, almost 55% of Sub-Saharan adults had access to financial accounts (Demirgüç-Kunt et al., 2022)². This assertion is further illustrated in Figure 2, highlighting the varying levels of financial inclusion across different regions. Given these economic and financial shifts, it becomes imperative to explore the role of financial inclusion in shaping the environment within the SSA landscape. Hence, we also empirically test the following null hypothesis: *There is no intra-regional heterogeneous influence of financial inclusion on environmental pollution levels in SSA*.

Concerning the economic impact of financial inclusion, existing studies suggest the presence of a non-linear inverted U-shaped relationship. The non-linear pattern aligns with the Environmental Kuznets Curve (EKC) hypothesis introduced by Grossman & Krueger (1995) in the context of economic development. According to the EKC hypothesis, environmental degradation occurs during the early development phase. Beyond a certain level of development, the economic benefit of growth becomes more affluent for environmental preservation. Though the EKC hypothesis is in the context of economic growth, it allows for relevant macroeconomic variables that are theorised to influence environmental sustainability (Qin et al., 2021), including financial inclusion (Qin et al., 2021; Renzhi & Baek, 2020; X. Wang et al., 2022). In the spirit of the non-linear relationship, we explore the EKC traditional theory within the context of financial inclusion. Our primary goal is not to test the traditional EKC hypothesis but to determine whether an adjusted model could be linked to the concept of financial inclusion. Hence, we empirically test the following null hypothesis: *There is no non-linear U-shaped relationship between financial inclusion and environmental pollution levels in SSA*.

We assessed how financial inclusion impacts green sustainability using a dynamic panel model based on the Resource Mobilisation Theory (McCarthy & Zald, 1977). We applied the System Generalized Methods of Moments (GMM) to test our hypothesis and account for potential endogeneity. We employed annual macro-level data from 2000 to 2021 using 38 SSA countries³. We used the Principal Component Analysis (PCA) to construct an environmental pollution index

¹ Third highest region in the world after Middle East (1.9%) and Asia Pacific (1.7%).

² From 2017 to 2021, the average rate of account ownership in developing economies increased by 8 percentage points, from 63 percent of adults to 71 percent of adults. In Sub-Saharan Africa, this expansion largely stems from the adoption of mobile money (Demirgüç-Kunt et al., 2022).

³ We do not perform estimation at micro-level as insights from such study would serve as further studies in a different paper with different research questions. This decision also stems from limitations in micro-level data accessibility, necessitating a different approach to data collection and analysis.

to measure Green Sustainability. Following the example of Opoku et al. (2022), the index captures four (4) environmental pollutants, including Particulate Matter (PM) 2.5, CO2 emission, methane emission, and nitrous oxide emission. We also employed renewable energy consumption as a measure of green sustainability. Our justification is based on empirical evidence highlighting the positive effect of renewable energy on air quality through controlling carbon emissions and lowering the direct interaction of households with toxic gases (Hanif, 2018). Renewable energy such as hydropower presents an appealing method for energy generation in addressing climate change, as they are promoted as a substitute for traditional fossil fuels (Castro-Diaz et al., 2023). We also employed a multidimensionality of financial inclusion to create a financial inclusion index. Our findings suggest that financial inclusion poses an unfavourable influence on the environment by facilitating the demand for more disposable goods and encouraging rapid expansion, leading to more waste generation and environmental pollution. On the other side, financial inclusion promotes the use of renewable energy.

Our research contributes significantly to the literature as it provides a novelty in the case of SSA. First, we assess how financial inclusion can be leveraged to create a sustainable environment and further combat climate change. As such, we include a composite measure of green sustainability, which has not been employed as a factor in previous research, and test a nonlinear relationship between financial inclusion and environmental pollution levels. Second, we provide a more focused analysis among the regions in SSA by conducting an intra-regional heterogeneity test to assess how this impact of financial inclusion on the environment differs among the various regions. Third, we provide more robust outcomes by controlling for innovative variables affecting the environment, such as urbanisation, trade openness, energy use, and human development. Fourth, we provide a clear insight into the magnitude of each financial inclusion indicator on the separate covariates of the environmental pollution index, as one indicator may affect the environment differently. The same insight is provided for separate pollutants that primarily impact climate change. Lastly, we provide policy recommendations on how financial inclusion could be applied by adopting innovative products like green loans, green bonds, and green credit score models to promote and strengthen the environment in SSA.

The paper is structured as follows: the next section describes the data and measurement methods, Section 3 provides the methodology and main estimations employed, Section 4 presents and discusses the main empirical results, section 5 presents the results of robustness tests and finally, Section 6 concludes and offers policy directions and suggestions for future research.

2. Data

We used panel data with annual frequencies covering 38 Sub-Saharan African (SSA) countries from 2000-2021 to investigate the role of financial inclusion on green sustainability, along with other country-level information detailed later in this section. The countries included in our study are listed in Appendix Table A1. The chosen period is motivated by the need to capture long-term trends, economic changes such as how financial inclusion has evolved and deepened to include more focus on sub-national data mostly since 2000, and environmental concerns in SSA, including the oil spills in Nigeria and the discovery of oil in Ghana. The data were sourced from the World Development Indicators database, the Global Financial Development Database and the Our World in Data. Appendix Tables A2, A3 and A4 list all the data used in our study.

2.1 Measure of Green Sustainability and Financial Inclusion

We measured green sustainability and financial inclusion by constructing an environmental pollution index (EPI) and a financial inclusion index (FII) to account for the several dimensions of both variables. Specific dimensions are defined with the help of relevant variables in Appendix Tables A3 and A4. We further introduced macro-level dependent variables to serve as controls.

2.1.1 Green sustainability measure

Following the example of Opoku et al. (2022), we measured green sustainability to depict the level of pollution intensity using an environmental pollution index. The index was constructed using four different environmental pollution variables, namely CO2 emissions, Particulate Matter 2.5 (PM), methane emissions, and nitrous oxide emissions. These variables were carefully selected to capture different facets of pollution, such as ecological footprint, carbon footprint, and air quality. We also avoided the issue of double-counting various pollution indicators.

CO2 emissions reflect the release of carbon oxide gas into the atmosphere primarily from activities like burning fossil fuels, solid waste, and trees. CO2 emission is measured in metric tons per capita, and by dividing the total emissions by population, it captures the contribution of the average citizen of each country. CO2 emissions have been associated with increased global warming and climate change concerns, making it a crucial indicator for evaluating environmental sustainability (You et al., 2015). Mitigating CO2 emissions is vital for curbing the adverse impacts of climate change, including extreme weather events and rising sea levels.

PM assesses the local level of air pollution and air quality. We employed PM as a variable to measure the concentration of small airborne particles, including dust, dirt, soot, smoke and liquid droplets. It primarily originates from various sources, including industrial processes, bushfires, pollens, sea spray, dust storms and motor vehicle engines (Isphording & Pestel, 2021). However, our empirical analysis focuses on PM 2.5 to measure particles up to a diameter of 2.5 microns. PM 2.5 is a commonly employed key indicator in literature to measure exposure to air pollution, and it stands as the fifth most significant risk factor contributing to global mortality (Cui et al., 2023). High levels of air pollution are detrimental to human health and the environment and can sometimes cause sleeplessness (Heyes & Zhu, 2019), making it an essential aspect of environmental sustainability.

Further, methane emission captures the emissions from agricultural activities, livestock farming, and waste management. Methane emission has implications for climate change due to the high heat-trapping ability of methane (CH4) compared to carbon dioxide. As is common in environmental research, the CH4 variable is measured as an equivalent of the metric tons of CO2 per capita. Finally, nitrous oxide emission captures emissions from agriculture and wastewater containing organic-based nitrogen materials. It is essential to note that not all types of nitrous emissions are considered. Instead, we focus specifically on one subset—nitrous oxide (N2O), which is a greenhouse gas with a longer atmospheric lifetime than other greenhouse gases. The primary source of N2O is the agricultural sector (Krüger & Tarach, 2022). While nitrous oxide is present in substantially smaller concentrations than carbon dioxide, nitrous oxide emission represents a much higher global warming potential (Paolini et al., 2018). Again, the N2O variable is measured in metric tons of CO2 equivalent per capita to provide a comparative perspective when assessing the impact of different greenhouse gases.

We also conducted a re-estimation using renewable energy as a single measure of green sustainability for robustness checks. Adopting renewable energy encourages a shift towards a more sustainable lifestyle and practice and the use of affordable and clean energy, as it does not produce additional greenhouse gases (Leroutier, 2022). It promotes energy efficiency, embraces eco-friendly technologies, combats climate change, and conserves natural resources. As such, using renewable energy as a measure helps gauge the reduction in greenhouse gas emissions and other pollutants, directly contributing to environmental sustainability.

As a further robustness check, we separated the pollutants that impact urban air quality (such as PM2.5) from those that primarily impact climate change (CO2, CH4, N2O). In order to

use the latter as a single measure, we constructed an index termed the climate footprint index (CFI). The equation for the index is described in Sub-section 2.1.3. We adopt this approach due to the distinct origins of our emissions variables—specifically, CO2, methane, and nitrous oxide emissions. These variables have been weighted based on their Global Warming Potential and differ slightly from PM2.5. Our approach is also to help gain insight into how financial inclusion impacts climate change in SSA. Finally, we performed another test examining the impact of financial inclusion on individual pollution indicators.

2.1.2 Financial inclusion measure

We assess financial inclusion by constructing a comprehensive Financial Inclusion Index (FII) comprising six variables. The index considers the multidimensional nature of financial inclusion, encompassing accessibility, usage, and the quality of financial services and products. Appendix Table A4 summarises the dimension defined with the help of relevant variables. Our rationale for this multifaceted approach is rooted in the understanding that true financial inclusion is not fully realised unless all these dimensions are effectively addressed. For instance, the mere presence of ATMs in proximity to individuals is unsatisfactory if those ATMs are non-functional or inaccessible (Nguyen, 2021).

To measure the accessibility dimension, we employed two key variables: ATMs per 100,000 adults and bank branches per 100,000 adults. These factors account for the physical availability of financial service points provided by financial institutions (Ugwuanyi et al., 2022; Cámara & Tuesta, 2014) and the country's banking infrastructure development.

As previously mentioned, the significance of financial access becomes apparent when it is actively utilised. Utilisation includes various financial activities such as saving, investing, borrowing, and making deposits and withdrawals. Building upon this concept, we utilised bank deposits and bank credits as indicators to quantify the extent of utilisation within the dimension of financial usage.

To measure the quality dimension, we employed two indicators: Life Insurance, captured as the ratio of life insurance premiums to a country's GDP, and Non-Life Insurance, measured as the ratio of non-life insurance premiums to a country's GDP. As denoted by the stated indicators, the quality dimension assesses the financial products or services that address financial needs and reduce financial burdens.

2.1.3 Principal Component Analysis

Further, we constructed the Environmental Pollution and Financial Inclusion Indexes using the Principal Component Analysis (PCA).⁴ The PCA method helps to gauge the aggregate effect of the four (4) and six (6) variables related to environmental pollution and dimensions of financial inclusion, respectively (Sharma and Changkakati, 2022). The application of PCA has three advantages: we avoid the problem of weight assignment and arbitrary weights using statistical weights (Cámara & Tuesta, 2014). PCA creates a linear combination of the original variables (principal components), which are orthogonal to each other. After the dimension reduction, statistical weights are applied to the various components based on the contribution of each principal component to the overall index. The weights are determined based on the eigenvalues of the covariance matrix of the standardised variables. We normalised the indices to ensure that all variables contribute equally to the analysis, preventing those with larger scales and variances from dominating the principal components. As such, the higher the eigenvalue, the more significant the variance and importance. We avoid the correlation between different individual indicators and can analyse the aggregate impact of the various factors without omitting any particular one (Iwasaki et al., 2022). To ensure coherent and consistent results, we normalise the various indicators to provide a comparable impact of green sustainability and financial inclusion independently of their original scale.

Hence, we employed the principal component methodology in line with the approach of Sharma and Changkakati (2022) and for every country i and period t, we developed an environmental pollution index (EPI) and a financial inclusion index (FII). We used the same approach to construct the Climate Footprint Index (CFI). All the indexes (*Index*_{it}) are defined in a similar way as follows:

$$Index_{it} = \sum_{n=1}^{N} \emptyset_n Z_{it} + \varepsilon_{it}, \tag{1}$$

where Z_{it} represents the dimensions associated with both financial services and products and also the dimensions of environmental pollution. The variable N represents the number of factors contributing to the index and the loadings \emptyset_n represents the respective weights for each principal

⁴ The PCA approach utilises the summation of weighted contributions from various dimensions associated with variables (in this case, financial services/products and environmental pollution) to create both indices. The use of the principal components ensures that the indices capture the most important variations in the underlying data.

component. The term ε_{it} accounts for errors in the model. The index is constructed so that an increase in financial inclusion increases emissions and, therefore, reduces green sustainability since the function of the index captures environmental pollutants.

The PCA results are presented in Appendix Table A5. The eigenvalues of the first two components explain the majority of variance in the data. The first two components of the environmental pollution and financial inclusion indexes account for 75% and 82% of the total variation, respectively. For the environmental pollution index, variables with higher absolute coefficients (Methane emission, nitrous oxide emission) have a stronger influence on the first principal component, indicating that changes in these variables contribute more to the variability captured by the first component. Variables with lower coefficients (PM25, CO2 emission) have a weaker influence on the first principal component. However, all variables are significant in determining environmental pollution levels.

All original variables have positive coefficients in the first component of the financial inclusion index, indicating that an increase in any of these variables contributes positively to the financial inclusion index.

2.2 Control Variables

Following the standard economic literature, we also accounted for other variables such as energy use (Usman et al., 2021; Sinha et al., 2017), human development⁵ (Opoku et al., 2022), natural resource dependency (Li et al., 2023), trade openness (Ali et al., 2016; Ozturk & Acaravci, 2013), economic growth (Ali et al., 2016; Usman et al., 2021), government expenditure (Pirgaip et al., 2023; Halkos & Paizanos, 2016), and urbanisation (Ali et al., 2016) as controls covariates.

We further present the descriptive statistics of variables in Appendix Table A6 and the results of our correlation analysis among variables in Appendix Table A7. We used the pairwise correlation test to determine the magnitude and linear direction between the variables, with -1 being a perfect negative correlation and +1 being a perfect correlation. Our results show no

⁵ We do not use the Human Development Index as a proxy for human development as it is highly correlated with urbanisation. We employ one of the single indicators of its composite dimensions to avoid the issue of multicollinearity.

evidence or concerns for multicollinearity in our model⁶. The independent variables have a correlation coefficient of less than 0.7 (Krehbiel, 2004).

3 Methodology

3.1 Model specification

We estimate the effect of financial inclusion on green sustainability through a dynamic panel model rooted in the Resource Mobilisation Theory. The model is based on the assumption that the present value of our selected variable, reflecting environmental pollution, exhibits a strong dependence on its own lagged values (Li et al., 2021). Pollution cumulates and generates damage over time with persistence (Calzolari et al., 2018). Consistent with Li et al. (2021), we adopt a dynamic panel model for individual i at time t of the form:

$$Y_{it} = \alpha + \sum_{s} \rho_{s} Y_{i,t-s} + \beta X_{it} + (u_{i} + \varepsilon_{it}); \quad (t = 2, 3, ..., T),$$
(2)

where $s = 1, 2, ..., and Y_{i,t-1}, Y_{i,t-2}$ are the values of the lagged dependent variable that affect X_{it} . The coefficient ρ_s captures the autocorrelation in Y, and β measures the effect of X on Y.

The lagged dependence stems from the fact that present environmental pollution levels are shaped by past practices, regulations, or environmental conditions and their likely evolution in the future (Sadorsky, 2010). This recognition emphasises the inherent link between environmental pollution or sustainability and its past values. We further estimate two equations to help test our hypothesis. Following the Resource Mobilisation Theory, and akin to Ozturk and Ullah (2022) and Sadorsky (2010), we break and specify our model assessing the impact of financial inclusion on environmental pollution as:

$$EPI_{it} = \beta_1 EPI_{it-1} + \beta_2 FII_{it} + \sum_{i=1}^n \beta_i Z_{it} + \nu_i + \varepsilon_{it}, \qquad (3)$$

In the above specification, the EPI represents the environmental pollution index, FII denotes the financial inclusion index, $\sum Z$ represents a vector of control variables, *v* represents the time-invariant country-specific fixed effects, and \mathcal{E} denotes the residual disturbance in the model estimation. The environmental pollution index and financial inclusion index encompass various specific dimensions, as detailed in Section 2.1, while Section 2.2 defines the control variables used.

⁶The Correlations between green sustainability, financial inclusion, energy use, human development, trade openness, natural resource, economic growth, government expenditure and urbanisation ranges between 0.02 and 0.53 and do not lead to problem of multi-collinearity.

We further modified the model to account for a potential non-linear relationship between financial inclusion and pollution. The adapted model incorporating this hypothesis is expressed as:

$$EPI_{it} = \beta_1 EPI_{it-1} + \beta_2 FII_{it} + \beta_3 FII_{it}^2 + \sum_{i=1}^n \beta_i Z_{it} + \nu_i + \varepsilon_{it},$$
(4)

Here, FII^2 captures the quadratic term of financial inclusion, accommodating the non-linear relationship postulated by the EKC hypothesis. Further, consistent with the mainstream literature, we modified our specification to include economic growth and its square as core components of the typical EKC model to assess the dynamics of financial inclusion's impact. We specify the model as:

$$EPI_{it} = \beta_1 EPI_{it-1} + \beta_2 FII_{it} + \beta_3 GDP_{it} + \beta_4 GDP_{it}^2 + \sum_{i=1}^n \beta_i Z_{it} + \nu_i + \varepsilon_{it},$$
(5)

where GDP and GDP² capture the linear and quadratic terms of economic growth in the EKC hypothesis, respectively.

In our robustness test, we re-estimated the model using the climate footprint index (CFI) as a proxy for our dependent variable. We specify the model as follows:

$$CFI_{it} = \beta_1 CFI_{it-1} + \beta_2 FII_{it} + \sum_{i=1}^n \beta_i Z_{it} + \nu_i + \varepsilon_{it},$$
(6)

where CFI represents the Carbon Footprint Index, as described in Sub-sections 2.1.1 and 2.1.3.

3.2 Estimation Technique

In line with the above model specifications (3) and (4) and consistent with existing literature, our dependent variable, green sustainability, a proxy by EPI, is considered endogenous. Endogeneity arises from the fact that green sustainability is influenced by its lag values, leading to a dynamic relationship (Law & Azman-Saini, 2012). Endogeneity can also occur in the dynamic model when there is a reverse causality among the independent variables and a reverse causality between the independent variable and financial inclusion (Francois, 2023).

We employ the System Generalized Methods of Moments (GMM) to estimate our results and address the endogeneity issue. The estimation strategy is efficient in handling endogeneity bias, particularly when appropriate instruments are available (Chen et al., 2024). The GMM technique, introduced by Blundell & Bond (1998) and Arellano & Bond (1991), is an instrumental variable method that combines a system of two equations, a model estimated in differences with a model estimated in levels for its estimation. Mathematically, the levels equation for computing the system GMM, according to Li et al. (2021), is computed as:

$$Y_{it} = aY_{it-t} + bX_{it} + yZ_{it} + \varepsilon_{it},$$
(7)

where Y_{it} is the dependent variable, X_{it} is the matrix of the exogenous variables, Z_{it} is the matrix of instruments (lagged levels and first differences of endogenous variables and exogenous variables), *a* and *b* are the parameters to be estimated, \mathcal{E}_{it} and is the error term.

Subsequently, the first difference equation is specified as:

$$Y_{it} = m\Delta Y_{it-1} + n\Delta X_{it} + q\Delta Z_{it} + \mu_{it},$$
(8)

where $\Delta Y_{it} = Y_{it} - Y_{it-1}$ is the first difference of the dependent variable; $\Delta X_{it} = X_{it} - X_{it-1}$ is the matrix of the first differences of exogenous variables; $\Delta Z_{it} = Z_{it} - Z_{it-1}$ is the matrix of the first differences of instruments; *m* and *n* are the parameters to be estimated, and μ_{it} and is the error term.

The System GMM uses moments conditions based on the instrumental variables. These moment conditions are used to construct the weighting matrix of the estimation. For each period t, the moment conditions are given by:

$$E[\Delta Z_{it} \cdot \mu_{it}] = 0, \tag{9}$$

where, for every t and i in the sample period, the expected value of the idiosyncratic error given the explanatory variables in all periods and the unobserved effect is 0 (Li et al., 2021).

Again, we adopted a panel estimation strategy as it is suitable in two (2) ways: (1) when the cross-sectional (N) units are more than the time series (T) units (Abeka et al., 2022); (2) when there is a tendency for persistence in the lagged differences of the dependent variable using a rule of thumb of 0.80 (Agyei et al., 2020). Thus, the coefficient of the lagged dependent variable should be close to 1.

In our study, T=22 and N=38. From our correlation analysis, the lag value of the dependent variable shows a correlation coefficient of 0.918, which is close to 1 and confirms the presence of lagged persistence. We employed the approach of Roodman (2009) to reduce the possibility of bias from instrument proliferation by restricting the moment conditions to a maximum of two lags

of the dependent variable. The number of instruments should be less than or equal to the number of groups to avoid instrument proliferation (David et al., 2023).

We assess the adequacy of our System GMM results and the validity of our instruments by reporting on the Sargan test of over-identifying restrictions (OIR) and Hansen test of over-identifying restrictions (OIR). Similarly, we test for the absence of autocorrelation using the Arellano and Bond serial correlation test. These tests require that the null hypothesis be rejected for the test to be valid. It is usually expected that the null hypothesis for the first order (AR1) will be rejected. Nevertheless, the second order (AR2) is more important because it will detect autocorrelation in levels and therefore, its null hypothesis should not be rejected. As Asongu and De Moor (2017) recommended, we further analyse and report on the difference-in-hansen test (DHT) for the exogeneity of instruments to assess the results of the Hansen OIR test. Fischer's test for the joint validity of estimated coefficients evaluates the validity of the estimated GMM model.

4. Results and Discussion

We present the results of our system GMM model, as detailed in Table 1, which examines the influence of financial inclusion on green sustainability. Our analysis shows that an increase in financial inclusion levels is strongly linked to increased pollution levels in SSA. The findings align with Le et al. (2020), which suggests that an increase in financial inclusion facilitates access to funds for acquiring more big-ticket items such as automobiles, refrigerators, air-conditioners, electricity generators and plants, whose use lies with fossil fuels and results in higher pollution. We further show that financial inclusion can lead to increased consumption of products due to increased purchasing power. While this can stimulate economic growth, it results in higher demand for disposal goods, leading to more waste generation and environmental pollution.

By comparing our results to existing literature, our finding aligns with Usman et al. (2021) in the Eurozone. Similarly, Popescu et al. (2023) confirm that Socially Responsible Investing (SRI) funds may not exhibit lower climate exposure compared to market indexes. They explained that funds labelled as sustainable (SRI) still maintain exposure to highly emitting companies, particularly those with high indirect carbon intensity. However, our results contrast with the findings of Shahbaz et al. (2022) and Renzhi and Baek (2020), who observed that financial inclusion reduces carbon emissions. These differing outcomes may be attributed to variations in the level of development of financial inclusion, industrialisation, technological advancement and innovation, and the preference for advanced gadgets in these countries. Nonetheless, in a region

like SSA, where financial inclusion is still in its early stages, a significant increase in financial inclusion creates the urge for rapid infrastructure development for growth, rapid urbanisation in search of economic opportunities, expansion of industries and increased demand for energy. These activities could destroy habitats and disrupt natural ecosystems, causing environmental harm.

4.1 Control for other macroeconomic determinants

We control for other determinants that can collaboratively influence environmental pollution by employing control variables of energy use, human development, natural resource dependency, trade openness, economic growth, government expenditure, and urbanisation. Our findings reveal a significant contribution of energy use (proxied by electricity from oil) in increasing environmental pollution levels. This adverse effect is evident in its negative coefficient and may be attributed to the extensive reliance on fossil fuels like oil for energy generation, transportation, and manufacturing (Le et al., 2020).

Further, we note that higher levels of human development significantly decrease environmental pollution levels. Human development improves awareness and behaviour change through education, contributing to adopting eco-friendly practices and policies. Our findings are consistent with Opoku, Dogah, et al. (2022).

Our analysis suggests that trade openness is linked to decreased environmental pollution levels. The presence of trade liberalisation encourages the alignment of environmental standards and policies with international norms, which fosters collective efforts to reduce pollution across borders. As noted by Le et al. (2020), trade agreements can enhance the capacity of governments to tackle environmental issues by eliminating trade barriers to environmental goods and services. Trade openness often facilitates technology and innovation transfer (Usman et al., 2021) to provide easier access to environmentally friendly technologies and renewable energy sources.

Also, increasing economic growth significantly contributes to high environmental pollution levels. Our result highlights how increased production of goods and services and overall economic expansion can contribute to environmental depletion. Rapid economic growth is associated with elevated energy consumption, intensified industrial activities, and heightened transportation, all of which contribute to environmental stress. In response to such adverse impacts from economic growth, Yuan et al. (2023) suggested adopting and establishing environmental courts, as seen in China. These specialised courts can reduce emissions without hindering economic growth. They achieve this by employing efficient judicial methods to address local

environmental pollution problems and establish a sustainable mechanism for environmental governance (Yuan et al., 2023).

We further highlight that government expenditure is linked with a decreasing impact on environmental pollution in SSA economies. An increasing number of government policies and regulations lead to increased government spending and efficient resource allocation, predominantly improving environmental efforts in SSA. According to Magacho et al. (2023), government expenditures are necessary to promote green industries either through direct fiscal stimulus or investment in renewable energy and other green technologies, which effectively reduces pollution. Moreover, many SSA countries are involved in international agreements, including the African Union Agenda 2063, the United Nations Framework Convention on Climate Change (UNFCCC), and the Paris Agreement. These agreements and intergovernmental environmental cooperation foster the advocacy of Clean Development Mechanism (CDM) (Cui et al. 2022) and encourage governments to invest in activities aimed at curbing greenhouse gas emissions, conserving biodiversity, and mitigating climate change.

Also, urbanisation contributes to increasing environmental deterioration in the sampled SSA economies. Urbanisation intensifies environmental effects, impacting air quality, ecosystems, land utilisation, natural cycles, water purity, waste handling, and the climate (Bai et al., 2017).

Finally, we show that the influence of natural resource dependence on environmental pollution is statistically irrelevant. The environmental impact of natural resource dependence may not be solely attributed to the region itself but may be dispersed globally due to the nature of the supply chain (OECD, 2023). The dispersion chain could account for the observed insignificant effect. Countries may export raw materials for processing elsewhere, influencing where pollution occurs in the supply chain. The insignificant effect may also stem from the varied environmental impacts of different natural resources (OECD, 2021). For instance, sustainable forestry may have less impact than oil or minerals. According to OECD (2019, 2021), the environmental impacts of other metals like aluminium, iron, and manganese are projected to remain constant or decrease over time due to the decarbonisation of energy in production and the increased use of secondary materials, which tend to have lower overall environmental impacts compared to primary materials.

4.2 Intra-regional heterogeneity with respect to environmental pollution

We further explored the intra-regional heterogeneous influence of financial inclusion on environmental pollution among the three regions in SSA: Western Africa, Eastern Africa and Southern Africa.⁷ Despite the recent increase in financial inclusion levels across SSA, disparities persist among the various regions. Specifically, Southern Africa exhibits relatively higher levels of financial inclusion than West Africa and East Africa (Sulemana & Dramani, 2022). Additionally, CO2 emission levels, as described in Figure 1, vary among the three regions. Hence, we examine and present the results of our intra-regional heterogeneity test in Table 2, where Eastern Africa is taken as a base and reported effects are measured with respect to it.

Table 2 results indicate a distinct intra-regional heterogeneity in SSA. First, we note that financial inclusion's impact on environmental pollution is 1.52% lower in Southern Africa than in Eastern Africa, which is consistent with the findings of Barut et al. (2023) for South Africa alone. Further, we show that given a 1% increase in financial inclusion in SSA, environmental pollution levels are 1.13% less in Western Africa than in Eastern Africa. Our finding aligns with Musah et al. (2023), showing that financial inclusion reduces environmental pollution in the Western African region. Based on the statistically significant coefficients, we can infer the presence of intra-regional heterogeneity within the Sub-Saharan Africa on environmental pollution is smaller than that of Eastern Africa. Hence, the two former regions seem to be in better shape concerning the environmental sustainability.

According to Wang et al. (2022), this heterogeneity could be attributed to varying levels of financial inclusion and regions' reliance on its advantageous geographical location to attract increased investments for environmental sustainability promotion. These results might also be due to these regional communities having varying policies, agreements, or initiatives contributing to environmental sustainability.

4.3 Test of a non-linear relationship

We also examined the existence of the non-linear relationship between financial inclusion and green sustainability in Sub-Saharan Africa. Our analysis in Table 1 revealed a distinctive inverted U-shaped pattern. Specifically, Table 1 shows a positive coefficient of financial inclusion on environmental pollution, indicating its contribution to increasing pollution levels. The effect is not straightforward, though. Since the coefficient of the quadratic term of financial inclusion is

⁷ The three regions represent countries that are members of the formal arrangements: the Economic Community of West African States (ECOWAS), the East African Community (EAC), and the Southern African Development Community (SADC). The last one is dubbed as Southern Africa and comprises sixteen members, including (the Republic of) South Africa.

negative and statistically significant, it confirms the presence of a non-linear U-shaped relationship between financial inclusion and environmental pollution in SSA.

These findings suggest that financial inclusion is linked with increasing environmental pollution. Further, the inverted U-shaped effect implies that in the early stages of increased financial inclusion, pollution level increases until it reaches an optimal level. Beyond the turning point, further increments in financial inclusion notably enhance sustainability. The U-shaped phenomenon can be explained by the behaviour of individuals and businesses during different stages of expansion of financial inclusion. Specifically, according to Renzhi and Baek (2020), people primarily focus on improving their living standards and expanding their businesses at the initial stages of increased financial inclusion. Such pursuit often leads to higher consumption and increased environmental degradation. However, at the later stage, individuals gain the capacity to invest in cleaner technologies, engage in more corporate social responsibilities, and explore greener innovations. Consequently, sustainability experiences a substantial improvement during these later stages. The shift aligns with Musah et al. (2023) and Youssef et al. (2020) and outlines the importance of reaching an optimal level of financial inclusion for achieving environmental sustainability in SSA.

Additionally, our result underscores the importance of considering financial inclusion and economic growth dynamics in fostering environmental sustainability. Our findings show that the quadratic term for GDP growth exhibits a negative impact, implying a curvilinear relationship between GDP growth and environmental pollution. Our result supports the typical Environmental Kuznets Curve (EKC) hypothesis (Li, 2023; Hlongwane & Daw, 2022) and serves as a battery test, indicating that our results align with the mainstream literature.

5. Robustness tests

In addition to our baseline estimation, we performed various robustness checks to verify the validity of our results. We explored alternative green sustainability and financial inclusion measures consistent with the approaches outlined in our base model. In Tables 3, 4, and 5, we present the results of our robustness tests, which are consistent and not materially different from our baseline results. Finally, we also re-estimated our baseline model and reported results based on alternative sets of instruments. We present the additional information in Appendix Table A8. The results show similarities with those reported in Tables 1 and 2. The probability values of the Sargan test indicate no rejection of the null hypothesis that the instruments as a group are

exogenous. Therefore, the instruments used in each GMM model estimation are valid with appropriate exclusion restrictions.

5.1 Robustness test I: Alternative measure of green sustainability

Firstly, we re-estimated the model using Renewable Energy (percentage of total renewable electricity output) as a proxy for green sustainability, which we term green energy production. Renewable electricity (mainly hydropower) constitutes SSA's most significant share of renewable energy supply (Abeka et al., 2022).

From the results in Table 3, our robustness analysis indicates a positive coefficient of the effect of financial inclusion on renewable energy production. The coefficient associated with financial inclusion shows that increasing financial inclusion levels is essential in promoting renewable energy production. In this respect, the impact of financial inclusion can be attributed to the substantial capital investment required to develop and implement renewable energy technologies and systems. Most households struggle to afford the upfront costs of installing clean energy systems (Harris, 2019). As such, the increased availability of financial resources contributes significantly to promoting, adopting, and expanding renewable energy production through broader initiatives. Our result is consistent with the findings of (Ofosu-Mensah Ababio et al., 2023; Li et al., 2022). Financial inclusion enables the renewable energy industry to access better financial investment, showing the essential characteristics of green finance as outlined by Zhou et al. (2023), which is necessary to develop green energy initiatives.

5.2 Robustness test II: Single alternative measures of financial Inclusion and green sustainability

We re-estimated our results using various individual measures of environmental pollution (CO2, N2O, CH4, PM 2.5) as a dependent variable and used various individual measures of financial inclusion as the regressors. Given the use of highly aggregated data, our robustness approach helps to identify the mechanisms through which financial inclusion impacts emissions and the magnitude of estimated effects. We present our results in Table 4. Our findings indicate that ATMs contribute to increased levels of environmental pollution in SSA. Their contribution is evidenced by a positive coefficient associated with nitrous oxide and methane emissions. This observation could be linked to increased economic activities enabled by improved financial accessibility, which can increase emission releases (Liu et al., 2022). The presence of bank branches is linked to a reduction in

nitrous oxide and methane emissions, but on the other hand, it contributes to an increase in CO2 emissions. This finding is consistent with Liu et al. (2022), who noted a small percentage impact of bank branches on carbon emission increase.

As seen from Table 4, bank credit enhances environmental sustainability by reducing the release of nitrous oxide and CO2 emissions across various activities. This result is intuitively correct since access to bank credit enables businesses to invest in clean energy projects, technologies and practices that may result in lower pollution (Zheng et al., 2023). Our results show that bank deposits are associated with lower pollution levels, as evidenced by negative coefficients with CO2, nitrous oxide, and methane emissions. Higher deposits could indicate a greater propensity for saving and a reduced need for immediate consumption, resulting in lower overall resource consumption and, subsequently, lower emissions. Moreover, banks may have substantial financing as deposits increase to support sustainable projects promoting environmental sustainability.

Further analysis highlights the positive impact of life insurance in reducing environmental pollution, particularly in mitigating methane and air pollution. The results suggest that individuals with life insurance coverage tend to adopt practices or behaviours leading to reduced pollution. Conversely, non-life insurance, mainly covering assets and properties, contributes to increased environmental pollution, as indicated by the positive coefficient on methane and PM2.5 emissions. We can infer that non-life insurance may indirectly incentivise behaviours or practices that lead to environmental pollution.

Our results suggest that ATMs and non-life insurance may hinder environmental sustainability within financial inclusion. These findings align with their impact on pollutants that primarily impact climate change (CO2, N2O, CH4) (as discussed in Sub-section 5.3), except for bank branches, which exhibit varying influences on specific pollution components. We further highlight that bank deposits, bank credit and life insurance play a beneficial role in reducing pollution levels.

5.3 Pls Robustness test III: Climate Footprint Index as a measure of green sustainability

Finally, as another measure of green sustainability, we focused on pollutants that primarily impact climate change (CO2, N2O, CH4) as a dependent variable. The results, outlined in Table 5, consistently support our initial findings, indicating that financial inclusion contributes to environmental pollution, as indicated by the climate footprint index. Specifically, we show that

ATMs, bank branches, and non-life insurance contribute to heightened levels of environmental pollution. The increase in these variables often signifies the expansion of industrialisation and infrastructure development, which jointly contribute to environmental pollution and the release of emissions through both direct and indirect mechanisms (Liu et al., 2022). On the other hand, bank credit, bank deposits, and life insurance play a vital role in reducing environmental pollution in the SSA. This observation is intuitively straightforward, as these variables are expected to provide credit to facilitate the adoption of green technologies and practices, and even reduce the long-term impact of environmental hazards.

Upon completing our robustness test, the results from the robustness analysis are not materially different from our baseline results. The results emphasise the reliability of our methodology, as they remain consistent even when alternative variables are employed to measure green sustainability and financial inclusion.

6. Conclusion and Policy Implications

We explored the relationship between financial inclusion and green sustainability in Africa and its link to environmental challenges. This relationship can have positive or negative effects. While it enables investments in green technologies and eco-friendly practices, it can also promote carbon emissions and energy-intensive consumption. In order to understand this relationship better, we developed three hypotheses to address the research objective better. We used a comprehensive empirical analysis spanning 38 Sub-Saharan African (SSA) countries from 2000 to 2021. We further developed an environmental pollution index and a financial inclusion index to account for the several dimensions of both variables. We also used the System GMM dynamic panel data estimation to address endogeneity and potential reverse causality issues.

Our results align with the notion that financial inclusion facilitates more accessible access to funds, fueling swift expansions in infrastructure, urbanisation, and increased energy consumption. However, this poses a significant risk of environmental degradation and a surge in waste generation. We found the presence of intra-regional heterogeneity within SSA. Southern Africa and Western Africa exhibit lower pollution levels than Eastern Africa. The variation suggests diverse regional strategies potentially influenced by varying financial inclusion and geographical factors. Our analysis's empirical findings also strongly support an inverted U-shaped relationship, consistent with the Environmental Kuznets Curve (EKC) hypothesis. We highlight the need to strike a balance in expanding financial services, emphasising the importance of reaching an optimal level of financial inclusion for fostering sustainable practices in the region. Perhaps we can conclude that the promotion of green sustainability tends to be more feasible for those with ample financial resources. Cultivating a culture of environmental awareness requires access to a diverse range of funds at an optimal level.

It is conventional to suggest that governments should continue promoting financial inclusion initiatives and encourage financial institutions to integrate environmental criteria into their lending policies and risk assessment processes. By incorporating sustainability considerations into their decision-making, financial institutions can incentivise businesses to adopt eco-friendly practices and prioritise investments in green technologies and projects. Such a policy may enable financial inclusion to be environmentally beneficial rather than unfavourable. It may involve providing incentives, subsidies, or reduced interest rates for loans and investments related to ecofriendly technologies like green bonds, green loans, green funds, and Sustainability-linked loans (SLLs). It is also worth noting that developing green credit score models that can evaluate financial institutions and their role in financing green sustainable projects compliance helps them get their commitment to financing and developing innovative products to guide investors toward genuinely eco-friendly initiatives. Lastly, policymakers are encouraged to develop long-term sustainability education and awareness programs tailored to different stages of financial inclusion. Recognising the U-shaped relationship between financial inclusion and green sustainability, they could focus on establishing basic environmental standards and gradually introduce more stringent regulations as financial inclusion progresses, ensuring that sustainable practices are integrated into business operations at an optimal stage of development. The financial market will also gradually respond to the opportunities embedded within these regulations. Strengthening environmental policies can potentially relocate capital towards green private sector investment, establishing a win-win benefit for all (Kruse et al., 2023).

Further research could assess the political and economic institutions that affect green sustainability in Sub-Saharan Africa and provide deeper insights into how certain factors, such as property rights and political stability, could impact green sustainability. Also, future research could assess this relationship at the micro-level to understand the behavioural aspects of individual financial inclusion levels and their impact on green sustainability.

References

- Abeka, M. J., Amoah, E. K., Owusu Appiah, M., Gatsi, J. G., Obuobi, N. K., & Boateng, E. (2022). Economic Institutions, Political Institutions and Renewable Energy Production in Africa. *Journal of African Business*, 23(4), 1049–1066. https://doi.org/10.1080/15228916.2021.1984818
- Agyei, S. K., Marfo-Yiadom, E., Ansong, A., & Idun, A. A. (2020). Corporate Tax Avoidance Incentives of Banks in Ghana. *Journal of African Business*, 21(4), 544–559. https://doi.org/10.1080/15228916.2019.1695183
- Ait Lahcen, M., & Gomis-Porqueras, P. (2021). A Model of Endogenous Financial Inclusion: Implications for Inequality and Monetary Policy. *Journal of Money, Credit and Banking*, 53(5), 1175–1209. https://doi.org/10.1111/jmcb.12808
- Ali, H. S., Law, S. H., & Zannah, T. I. (2016). Dynamic impact of urbanization, economic growth, energy consumption, and trade openness on CO 2 emissions in Nigeria. *Environmental Science* and Pollution Research, 23(12), 12435–12443. https://doi.org/10.1007/s11356-016-6437-3
- Arellano, M., & Bond, S. (1991). Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations. *The Review of Economic Studies*, 58(2), 277. https://doi.org/10.2307/2297968
- Asafu-Adjaye, J. (2014). The Economic Impacts of Climate Change on Agriculture in Africa. *Journal* of African Economies, 23(suppl 2), ii17–ii49. https://doi.org/10.1093/jae/eju011
- Bai, X., McPhearson, T., Cleugh, H., Nagendra, H., Tong, X., Zhu, T., & Zhu, Y.-G. (2017). Linking Urbanization and the Environment: Conceptual and Empirical Advances. *Annual Review of Environment and Resources*, 42(1), 215–240. https://doi.org/10.1146/annurev-environ-102016-061128
- Barut, A., Kaya, E., Bekun, F. V., & Cengiz, S. (2023). Environmental sustainability amidst financial inclusion in five fragile economies: Evidence from lens of environmental Kuznets curve. *Energy*, 269, 126802. https://doi.org/10.1016/j.energy.2023.126802
- Beck, T. (2020). *Fintech and Financial Inclusion: Opportunities and Pitfalls (ADBI Working Paper No.* 1165). Asian Development Bank Institute. https://www.adb.org/sites/default/files/publication/623276/adbi-wp1165.pdf
- Blundell, R., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics*, 87(1), 115–143. https://doi.org/10.1016/S0304-4076(98)00009-8
- Calzolari, G., Casari, M., & Ghidoni, R. (2018). Carbon is forever: A climate change experiment on cooperation. *Journal of Environmental Economics and Management*, 92, 169–184. https://doi.org/10.1016/j.jeem.2018.09.002
- Cámara, N., & Tuesta, D. (2014). Measuring Financial Inclusion: A Muldimensional Index. *BBVA Research Paper No. 14/26*. https://doi.org/10.2139/ssrn.2634616
- Castro-Diaz, L., García, M. A., Villamayor-Tomas, S., & Lopez, M. C. (2023). Impacts of hydropower development on locals' livelihoods in the Global South. *World Development*, 169, 106285. https://doi.org/10.1016/j.worlddev.2023.106285
- Chen, D., Isah, A. B., & Gummi, U. M. (2024). Effect of Foreign Direct Investments on Industrialisation: The Case of China in Africa. *Journal of African Economies*, ejad026. https://doi.org/10.1093/jae/ejad026
- Cui, J., Huang, S., & Wang, C. (2023). The impact of air quality on innovation activities in China. Journal of Environmental Economics and Management, 122, 102893. https://doi.org/10.1016/j.jeem.2023.102893
- Cui, J., Wang, Z., & Yu, H. (2022). Can International Climate Cooperation Induce Knowledge Spillover to Developing Countries? Evidence from CDM. *Environmental and Resource Economics*, 82(4), 923–951. https://doi.org/10.1007/s10640-022-00697-8

- Dai, X., Qian, S., & Zhang, J. (2022). Sustainable financial inclusion as a source of green environment? Evidence from selected regional comprehensive economic partnership countries. *Economic Research-Ekonomska Istraživanja*, *35*(1), 5719–5738. https://doi.org/10.1080/1331677X.2022.2035244
- David, A., Diallo, Y., & Nilsson, B. (2023). Informality and Inequality: The African Case. *Journal of African Economies*, 32(Supplement_2), ii273–ii295. https://doi.org/10.1093/jae/ejac052
- Demirgüç-Kunt, A., Klapper, L., Singer, D., & Ansar, S. (2022). The Global Findex Database 2021: Financial Inclusion, Digital Payments, and Resilience in the Age of COVID-19. The World Bank. https://doi.org/10.1596/978-1-4648-1897-4
- Dou, D., & Li, L. (2022). Does sustainable financial inclusion and energy efficiency ensure green environment? Evidence from B.R.I.C.S. countries. *Economic Research-Ekonomska Istraživanja*, 35(1), 5599–5614. https://doi.org/10.1080/1331677X.2022.2032785
- Energy Institute. (2023). Statistical Review of the World Energy (72nd ed.). Energy Institute.
- Francois, J. N. (2023). Habits, Rule-of-thumb Consumption and Useful Public Consumption in Sub-Sahara Africa: Theory and New Evidence. *Journal of African Economies*, *32*(5), 469–494. https://doi.org/10.1093/jae/ejac024
- Frankel, J. A., & Romer, D. (1999). Does Trade Cause Growth? *American Economic Review*, 89(3), 379–399. https://doi.org/10.1257/aer.89.3.379
- Grossman, G. M., & Krueger, A. B. (1995). Economic Growth and the Environment. *The Quarterly Journal of Economics*, 110(2), 353–377. https://doi.org/10.2307/2118443
- Hafeez, M., Yuan, C., Shahzad, K., Aziz, B., Iqbal, K., & Raza, S. (2019). An empirical evaluation of financial development-carbon footprint nexus in One Belt and Road region. *Environmental Science and Pollution Research*, 26(24), 25026–25036. https://doi.org/10.1007/s11356-019-05757-z
- Halkos, G. E., & Paizanos, E. A. (2016). The effects of fiscal policy on CO2 emissions: Evidence from the U.S.A. *Energy Policy*, 88, 317–328. https://doi.org/10.1016/j.enpol.2015.10.035
- Hanif, I. (2018). Impact of economic growth, nonrenewable and renewable energy consumption, and urbanization on carbon emissions in Sub-Saharan Africa. *Environmental Science and Pollution Research*, 25(15), 15057–15067. https://doi.org/10.1007/s11356-018-1753-4
- Harris, L. (2019). Beyond the Grid: Engineering, Institutions and Finance. Journal of African Economies, 28(6), i41–i50. https://doi.org/10.1093/jae/ejz024
- Heyes, A., & Zhu, M. (2019). Air pollution as a cause of sleeplessness: Social media evidence from a panel of Chinese cities. *Journal of Environmental Economics and Management*, 98, 102247. https://doi.org/10.1016/j.jeem.2019.07.002
- Hlongwane, N. W., & Daw, O. D. (2022). Testing Environmental Kuznets Curve Hold in South Africa: An Econometric Approach. *International Journal of Energy Economics and Policy*, 12(3), 385–394. https://doi.org/10.32479/ijeep.12875
- Hsu, C.-C., Chau, K. Y., & Chien, F. (2023). Natural resource volatility and financial development during Covid-19: Implications for economic recovery. *Resources Policy*, 81, 103343. https://doi.org/10.1016/j.resourpol.2023.103343
- Hussain, S., Ahmad, T., & Shahzad, S. J. H. (2021). Financial Inclusion and CO2 Emissions in Asia: Implications for Environmental Sustainability [Preprint]. In Review. https://doi.org/10.21203/rs.3.rs-245990/v1
- Isphording, I. E., & Pestel, N. (2021). Pandemic meets pollution: Poor air quality increases deaths by COVID-19. *Journal of Environmental Economics and Management*, *108*, 102448. https://doi.org/10.1016/j.jeem.2021.102448
- Iwasaki, I., Kočenda, E., & Shida, Y. (2022). Institutions, financial development, and small business survival: Evidence from European emerging markets. *Small Business Economics*, 58(3), 1261– 1283. https://doi.org/10.1007/s11187-021-00470-z

- Jensen, A. L. (1996). Beverton and Holt life history invariants result from optimal trade-off of reproduction and survival. *Canadian Journal of Fisheries and Aquatic Sciences*, 53(4), 820–822. https://doi.org/10.1139/f95-233
- Ji, X., Chen, X., Mirza, N., & Umar, M. (2021). Sustainable energy goals and investment premium: Evidence from renewable and conventional equity mutual funds in the Euro zone. *Resources Policy*, 74, 102387. https://doi.org/10.1016/j.resourpol.2021.102387
- Ji, X., Umar, M., Ali, S., Ali, W., Tang, K., & Khan, Z. (2021). Does fiscal decentralization and ECO-INNOVATION promote sustainable environment? A case study of selected fiscally decentralized countries. *Sustainable Development*, 29(1), 79–88. https://doi.org/10.1002/sd.2132
- Krehbiel, T. C. (2004). Correlation Coefficient Rule of Thumb. *Decision Sciences Journal of Innovative Education*, 2(1), 97–100. https://doi.org/10.1111/j.0011-7315.2004.00025.x
- Krüger, J. J., & Tarach, M. (2022). Greenhouse Gas Emission Reduction Potentials in Europe by Sector: A Bootstrap-Based Nonparametric Efficiency Analysis. *Environmental and Resource Economics*, 81(4), 867–898. https://doi.org/10.1007/s10640-022-00660-7
- Kruse, T., Mohnen, M., & Sato, M. (2023). Do Financial Markets Respond to Green Opportunities? Journal of the Association of Environmental and Resource Economists, 727370. https://doi.org/10.1086/727370
- Law, S. H., & Azman-Saini, W. N. W. (2012). Institutional quality, governance, and financial development. *Economics of Governance*, 13(3), 217–236. https://doi.org/10.1007/s10101-012-0112-z
- Le, T.-H., Le, H.-C., & Taghizadeh-Hesary, F. (2020). Does financial inclusion impact CO2 emissions? Evidence from Asia. *Finance Research Letters*, *34*, 101451. https://doi.org/10.1016/j.frl.2020.101451
- Leroutier, M. (2022). Carbon pricing and power sector decarbonization: Evidence from the UK. Journal of Environmental Economics and Management, 111, 102580. https://doi.org/10.1016/j.jeem.2021.102580
- Lessmann, K., & Kalkuhl, M. (2023). Climate finance intermediation: Interest spread effects in a climate policy model. *Journal of the Association of Environmental and Resource Economists*, 725920. https://doi.org/10.1086/725920
- Li, J. (2023). Environmental Kuznets curve, balanced growth, and influencing factors: Evidence from economic development in China. *International Journal of Climate Change Strategies and Management*. https://doi.org/10.1108/IJCCSM-08-2022-0116
- Li, J., Ding, H., Hu, Y., & Wan, G. (2021). Dealing with dynamic endogeneity in international business research. *Journal of International Business Studies*, 52(3), 339–362. https://doi.org/10.1057/s41267-020-00398-8
- Li, J., Dong, K., Wang, K., & Dong, X. (2023). How does natural resource dependence influence carbon emissions? The role of environmental regulation. *Resources Policy*, 80, 103268. https://doi.org/10.1016/j.resourpol.2022.103268
- Li, J., Dong, X., & Dong, K. (2022). How much does financial inclusion contribute to renewable energy growth? Ways to realize green finance in China. *Renewable Energy*, 198, 760–771. https://doi.org/10.1016/j.renene.2022.08.097
- Liu, D., Zhang, Y., Hafeez, M., & Ullah, S. (2022). Financial inclusion and its influence on economicenvironmental performance: Demand and supply perspectives. *Environmental Science and Pollution Research*, 29(38), 58212–58221. https://doi.org/10.1007/s11356-022-18856-1
- Magacho, G., Espagne, E., Godin, A., Mantes, A., & Yilmaz, D. (2023). Macroeconomic exposure of developing economies to low-carbon transition. *World Development*, 167, 106231. https://doi.org/10.1016/j.worlddev.2023.106231
- McCarthy, J. D., & Zald, M. N. (1977). Resource Mobilization and Social Movements: A Partial Theory. *American Journal of Sociology*, 82(6), 1212–1241.

- Melo, J. de, & Solleder, J.-M. (2022). *The Landscape of CO2 Emissions Across Africa: A Comparative Perspective*. https://hal.science/hal-03739898
- Musah, M., Gyamfi, B. A., Kwakwa, P. A., & Agozie, D. Q. (2023). Realizing the 2050 Paris climate agreement in West Africa: The role of financial inclusion and green investments. *Journal of Environmental Management*, 340, 117911. https://doi.org/10.1016/j.jenvman.2023.117911
- Nguyen, T. T. H. (2021). Measuring financial inclusion: A composite FI index for the developing countries. *Journal of Economics and Development*, 23(1), 77–99. https://doi.org/10.1108/JED-03-2020-0027
- OECD. (2019). OECD Global Material Resources Outlook to 2060. OECD Environment Statistics (database).
- OECD. (2021). Towards a More Resource-Efficient and Circular Economy. OECD.
- OECD. (2023). Handbook on Environmental Due Diligence in Mineral Supply Chains. OECD. https://doi.org/10.1787/cef843bf-en
- Ofosu-Mensah Ababio, J., Yiadom, E. B., Mawutor, J. K. M., Tuffour, J. K., & Attah-Botchwey, E. (2023). Sustainable energy for all: The link between financial inclusion, renewable energy and environmental sustainability in developing economies. *International Journal of Energy Sector Management*. https://doi.org/10.1108/IJESM-07-2023-0012
- Opoku, E. E. O., Acheampong, A. O., Dzator, J., & Kufuor, N. K. (2022). Does environmental sustainability attract foreign investment? Evidence from developing countries. *Business Strategy and the Environment*, *31*(7), 3542–3573. https://doi.org/10.1002/bse.3104
- Opoku, E. E. O., Dogah, K. E., & Aluko, O. A. (2022). The contribution of human development towards environmental sustainability. *Energy Economics*, 106, 105782. https://doi.org/10.1016/j.eneco.2021.105782
- Ozturk, I., & Acaravci, A. (2013). The long-run and causal analysis of energy, growth, openness and financial development on carbon emissions in Turkey. *Energy Economics*, *36*, 262–267. https://doi.org/10.1016/j.eneco.2012.08.025
- Ozturk, I., & Ullah, S. (2022). Does digital financial inclusion matter for economic growth and environmental sustainability in OBRI economies? An empirical analysis. *Resources, Conservation and Recycling*, *185*, 106489. https://doi.org/10.1016/j.resconrec.2022.106489
- Paolini, V., Petracchini, F., Segreto, M., Tomassetti, L., Naja, N., & Cecinato, A. (2018). Environmental impact of biogas: A short review of current knowledge. *Journal of Environmental Science and Health, Part A*, 53(10), 899–906. https://doi.org/10.1080/10934529.2018.1459076
- Pirgaip, B., Bayrakdar, S., & Kaya, M. V. (2023). The role of government spending within the environmental Kuznets curve framework: Evidence from G7 countries. *Environmental Science* and Pollution Research, 30(34), 81513–81530. https://doi.org/10.1007/s11356-023-25180-9
- Popescu, I.-S., Gibon, T., Hitaj, C., Rubin, M., & Benetto, E. (2023). Are SRI funds financing carbon emissions? An input-output life cycle assessment of investment funds. *Ecological Economics*, 212, 107918. https://doi.org/10.1016/j.ecolecon.2023.107918
- Qin, L., Raheem, S., Murshed, M., Miao, X., Khan, Z., & Kirikkaleli, D. (2021). Does financial inclusion limit carbon dioxide emissions? Analyzing the role of globalization and renewable electricity output. *Sustainable Development*, 29(6), 1138–1154. https://doi.org/10.1002/sd.2208
- Renzhi, N., & Baek, Y. J. (2020). Can financial inclusion be an effective mitigation measure? Evidence from panel data analysis of the environmental Kuznets curve. *Finance Research Letters*, 37, 101725. https://doi.org/10.1016/j.frl.2020.101725
- Roodman, D. (2009). How to do Xtabond2: An Introduction to Difference and System GMM in Stata. *The Stata Journal: Promoting Communications on Statistics and Stata*, 9(1), 86–136. https://doi.org/10.1177/1536867X0900900106

- Sadorsky, P. (2010). The impact of financial development on energy consumption in emerging economies. *Energy Policy*, *38*(5), 2528–2535. https://doi.org/10.1016/j.enpol.2009.12.048
- Shahbaz, M., Li, J., Dong, X., & Dong, K. (2022). How financial inclusion affects the collaborative reduction of pollutant and carbon emissions: The case of China. *Energy Economics*, 107, 105847. https://doi.org/10.1016/j.eneco.2022.105847
- Sharma, U., & Changkakati, B. (2022). Dimensions of global financial inclusion and their impact on the achievement of the United Nations Development Goals. *Borsa Istanbul Review*, 22(6), 1238–1250. https://doi.org/10.1016/j.bir.2022.08.010
- Sinha, A., Shahbaz, M., & Balsalobre, D. (2017). Exploring the relationship between energy usage segregation and environmental degradation in N-11 countries. *Journal of Cleaner Production*, 168, 1217–1229. https://doi.org/10.1016/j.jclepro.2017.09.071
- Sulemana, M., & Dramani, J. B. (2022). Effect of financial sector development and institutions on economic growth in SSA. Does the peculiarities of regional blocs matter? *Journal of Sustainable Finance & Investment*, 12(4), 1102–1124. https://doi.org/10.1080/20430795.2020.1837500
- Udeagha, M. C., & Muchapondwa, E. (2022). Investigating the moderating role of economic policy uncertainty in environmental Kuznets curve for South Africa: Evidence from the novel dynamic ARDL simulations approach. *Environmental Science and Pollution Research*, 29(51), 77199–77237. https://doi.org/10.1007/s11356-022-21107-y
- Udeagha, M. C., & Muchapondwa, E. (2023). Green finance, fintech, and environmental sustainability: Fresh policy insights from the BRICS nations. *International Journal of Sustainable Development* & *World* Ecology, 30(6), 633–649. https://doi.org/10.1080/13504509.2023.2183526
- Ugwuanyi, U., Ugwuoke, R., Onyeanu, E., Festus Eze, E., Isahaku Prince, A., Anago, J., & Ibe, G. I. (2022). Financial inclusion economic growth nexus: Traditional finance versus digital finance in Sub-Saharan Africa. *Cogent Economics & Finance*, 10(1), 2133356. https://doi.org/10.1080/23322039.2022.2133356
- Usman, M., Makhdum, M. S. A., & Kousar, R. (2021). Does financial inclusion, renewable and nonrenewable energy utilization accelerate ecological footprints and economic growth? Fresh evidence from 15 highest emitting countries. *Sustainable Cities and Society*, 65, 102590. https://doi.org/10.1016/j.scs.2020.102590
- Wang, L., Wang, Y., Sun, Y., Han, K., & Chen, Y. (2022). Financial inclusion and green economic efficiency: Evidence from China. *Journal of Environmental Planning and Management*, 65(2), 240–271. https://doi.org/10.1080/09640568.2021.1881459
- Wang, X., Wang, X., Ren, X., & Wen, F. (2022). Can digital financial inclusion affect CO2 emissions of China at the prefecture level? Evidence from a spatial econometric approach. *Energy Economics*, 109, 105966. https://doi.org/10.1016/j.eneco.2022.105966
- You, W.-H., Zhu, H.-M., Yu, K., & Peng, C. (2015). Democracy, Financial Openness, and Global Carbon Dioxide Emissions: Heterogeneity Across Existing Emission Levels. World Development, 66, 189–207. https://doi.org/10.1016/j.worlddev.2014.08.013
- Youssef, A. B., Boubaker, S., & Omri, A. (2020). Financial development and macroeconomic sustainability: Modeling based on a modified environmental Kuznets curve. *Climatic Change*, 163(2), 767–785. https://doi.org/10.1007/s10584-020-02914-z
- Yuan, H., Zou, L., & Feng, Y. (2023). How to achieve emission reduction without hindering economic growth? The role of judicial quality. *Ecological Economics*, 209, 107839. https://doi.org/10.1016/j.ecolecon.2023.107839
- Zaidi, S. A. H., Hussain, M., & Uz Zaman, Q. (2021). Dynamic linkages between financial inclusion and carbon emissions: Evidence from selected OECD countries. *Resources, Environment and Sustainability*, 4, 100022. https://doi.org/10.1016/j.resenv.2021.100022

- Zheng, S., Sheng, B., Ghafoor, A., Ashraf, A. A., & Qamri, G. M. (2023). Investigating the environmental externalities of digital financial inclusion and the COVID-19 pandemic: An environmental sustainability perspective. *Environmental Science and Pollution Research*, 30(33), 80758–80767. https://doi.org/10.1007/s11356-023-27433-z
- Zhou, W., Wu, X., & Zhou, D. (2023). Does green finance reduce environmental pollution?—A study based on China's provincial panel data. *Environmental Science and Pollution Research*, 30(59), 123862–123881. https://doi.org/10.1007/s11356-023-30738-8

Tables

LINEAR **NON-LINEAR NON-LINEAR** (GDP) (FII) Lag of Environmental Pollution Index 0.948*** 0.898*** 0.956*** (0.000)(0.000)(0.000)0.052*** 0.049*** **Financial Inclusion** 0.125*** (0.004)(0.000)(0.009)-0.019*** Financial Inclusion_squared (0.000)0.014*** Economic Growth (0.000)-0.004*** Economic Growth squared (0.000)Controls 0.050*** Energy Use 0.065** 0.061*** (0.008)(0.017)(0.002)-0.011*** -0.009*** Human Development 0.001 (0.000)(0.753)(0.001)Natural Resources 0.004* 0.001 0.004*(0.062)(0.311)(0.067)-0.005*** **Trade Openness** -0.001** -0.000 (0.047)(0.000)(0.269)0.009*** 0.004*** Economic Growth (0.001)(0.014)Government Expenditure -0.023*** 0.031*** -0.024*** (0.002)(0.000)(0.000)0.005*** Urbanization 0.004** 0.002 (0.242)(0.029)(0.002)0.368*** Constant 0.289*** -0.444*** (0.001)(0.000)(0.000)Diagnostics Wald Test 4280.39 27225.80 14173.76 Prob. (Wald) 0.000 0.000 0.000 AR1 (p-value) 0.231 0.227 0.224 AR2 (p-value) 0.365 0.363 0.363 Hansen-J (p-value) 0.548 0.592 0.316 Sargan (p-value) 0.439 0.347 0.285 DHT for instrument a. Instruments in levels H excluding group 0.242 0.625 0.267 b. IV (years, eq(diff)) H excluding group 0.706 0.312 0.41 Number of groups 33 33 33 Number of instrumental 30 30 30 162 Observations 162 162

Table 1: Effect of Financial Inclusion on Environmental Pollution

Model 2 uses the financial inclusion index as a test of a non-linear pattern, while Model 3 utilizes GDP as a battery test of the typical EKC hypothesis. Statistical significance at the 1%, 5%, and 10% level is indicated by ***, **, and *, respectively

	INTRA-REGIONAL
Lag of Environmental Pollution Index	0.720***
	(0.000)
Financial Inclusion in Southern Africa	-1.515***
	(0.000)
Financial Inclusion in Western Africa	-1.133***
	(0.000)
Controls	(((((((((((((((((((((((((((((((((((((((
Energy Use	-0.076***
	(0.001)
Human Development	-0.018***
*	(0.007)
Natural Resources	-0.009***
	(0.003)
Trade Openness	-0.001
*	(0.391)
Economic Growth	0.033***
	(0.000)
Government Expenditure	0.024***
-	(0.008)
Urbanization	-0.003
	(0.541)
Constant	0.982***
	(0.000)
Diagnostics	
Wald Test	3544.37
Prob. (Wald)	0.000
AR1 (p-value)	0.217
AR2 (p-value)	0.368
Hansen-J (p-value)	0.338
Sargan (p-value)	0.252
DHT for instrument	
a. Instruments in levels	
H excluding group	
b. IV (years, eq(diff))	
H excluding group	0.501
Number of groups	33
Number of instrumental	30
Observations	162

Table 2: Intra-regional heterogeneity with respect to environmental pollution

Robustness Tests

	RENEWABLE ENERGY
Lag of Renewable Energy	0.996***
	(0.000)
Financial Inclusion	1.112***
	(0.000)
Controls	
Energy Use	0.852*
	(0.069)
Human Development	0.272***
	(0.001)
Natural Resources	0.379***
	(0.000)
Trade Openness	0.049**
	(0.029)
Economic Growth	-0.454
	(0.000)
Government Expenditure	-0.347***
	(0.010)
Urbanization	-0.300***
	(0.001)
Constant	7.615***
	(0.004)
Diagnostics	
Wald Test	59401.87
Prob. (Wald)	0.000
AR1 (p-value)	0.045
AR2 (p-value)	0.337
Hansen-J (p-value)	0.260
Sargan (p-value)	0.754
DHT for instrument	
a. Instruments in levels	
H excluding group	0.244
b. IV (years, eq(diff))	
H excluding group	0.226
Number of groups	30
Number of instrumental	33
Observations	217

Table 3: Robustness Test I - Alternative measure of green sustainability

¥	CO2 EMISSION	N2O	METHANE	PM2.5
· · · · · · · · · · · · · · · · · · ·	0.046555	EMISSION	0.045444	0.500
Lag of environmental pollutants	0.946***	0.767***	0.847/***	0.789
	(0.000)	(0.000)	(0.000)	(0.000)
ATMs	-0.002*	0.003**	0.006***	-0.044
	(0.091)	(0.018)	(0.000)	(0.258)
Bank Branches	0.013***	-0.014**	-0.013***	0.231
	(0.000)	(0.028)	(0.001)	(0.290)
Bank Credit to Deposit	-0.002***	-0.002**	-0.002	0.004
	(0.000)	(0.034)	(0.124)	(0.941)
Bank Deposit	-0.005***	-0.005**	-0.006***	-0.158
	(0.000)	(0.024)	(0.001)	(0.233)
Life Insurance	0.035***	-0.043	-0.076***	-3.461**
	(0.007)	(0.149)	(0.000)	(0.046)
Non-Life Insurance	0.012	0.120	0.182***	9.819***
	(0.719)	(0.254)	(0.004)	(0.009)
Controls			()	(/
Energy Use	0.047***	0.055**	0.038***	1.557
	(0.000)	(0.048)	(0.007)	(190)
Human Development	0.010***	0.005	0.005*	0.438**
	(0,000)	(0.127)	(0.058)	(0.034)
Natural Resources	0.001	-0.004*	-0.011***	0 269***
Nutural Resources	(0.436)	(0.062)	(0.001)	(0.009)
Trade Openness	0.002**	0.002)	0.001)	(0.00)
Trade Openness	(0.034)	(0.002)	(0,004)	(0.034)
Economia Crowth	(0.034)	(0.031)	(0.004)	(0.034)
Economic Growin	(0.002^{44})	(0.013^{+++})	(0.002^{+++})	-0.122
Concernant Errora distant	(0.024)	(0.009)	(0.000)	(0.439)
Government Expenditure	-0.003	-0.011****	-0.012****	0.091*
TT 1	(0.506)	(0.002)	(0.001)	(0.089)
Urbanization	-0.004***	0.003	0.005**	-0.18/*
	(0.000)	(0.225)	(0.017)	(0.069)
Constant	0.188***	0.232	0.028	7.289
	(0.000)	(0.117)	(0.882)	(0.127)
Diagnostics				
Wald Test	229058.02	22100.00	51748.76	2585.89
Prob. (Wald)	0.000	0.000	0.000	0.000
AR1 (p-value)	0.068	0.273	0.272	0.041
AR2 (p-value)	0.559	0.304	0.302	0.110
Hansen-J (p-value)	0.790	0.864	0.454	0.937
Sargan (p-value)	0.218	0.124	0.109	0.966
DHT for instrument				
a. Instruments in levels				
H excluding group	0.279	0.512	0.601	0.372
b. IV (years, eq(diff))				
H excluding group	0.755	0.911	0.5811	0.918
Number of groups	34	34	34	33
Number of instruments	30	34	34	30
Observations	289	289	289	162

Table 4: Robustness	Test II -	• Single	alternative	measures	of	financial	inclusion	and	green
sustainability									-

	CLIMATE FOOTPRINT INDEX (1)	CLIMATE FOOTPRINT INDEX (2)
Lag of Climate Change Index	0.763***	0.677***
	0.000	(0.000)
Financial Inclusion	0.071***	(
	(0.000)	
ATMs	(0.000)	0.005***
		(0.006)
Bank Branches		0.023**
		(0.025)
Bank Credit to Deposit		-0.007***
Durin Credit to Deposit		(0,009)
Bank Deposit		-0.019***
Builly Deposit		(0.003)
Life Insurance		-0.271***
		(0,000)
Non-Life Insurance		0.654***
Non-Life insurance		(0,000)
Controls		(0.000)
Energy Use	0.044***	0 132***
Energy Ose	(0,000)	(0.007)
Human Development	0.013***	0.018
Human Development	(0,000)	(0.141)
Natural Pasouroos	0.015***	(0.141)
Natural Resources	(0,000)	(0.040)
Trada Ononnaga	(0.000)	(0.040)
Trade Openness	(0,000)	(0.207)
Essenceric Crowth	(0.000)	(0.297)
Economic Growth	(0.023****	(0.021^{++++})
Communent Errore ditara	(0.000)	(0.011)
Government Expenditure	-0.04/*****	-0.048
TT 1 · /	(0.000)	(0.001)
Urbanization	-0.001	-0.008
Constant	(0.609)	(0.270)
Constant	0.580***	1.182^{***}
	(0.000)	(0.010)
Diagnostics	20550.00	20400.00
Wald Test	20558.89	30400.00
Prob. (Wald)	0.000	0.000
ARI (p-value)	0.279	0.262
AR2 (p-value)	0.305	0.304
Hansen-J (p-value)	0.847	0.534
Sargan (p-value)	0.183	0.238
DHT for instrument		
a. Instruments in levels	0.401	0.054
H excluding group	0.401	0.254
b. IV (years, eq(diff))	0.075	0. (22
H excluding group	0.875	0.623
Number of groups	34	34
Number of instruments	34	34
Observations	289	289

Table 5: Robustness Test III - Climate Footprint Index as a measure of green sustainability CLIMATE FOOTPRINT INDEX (1) CLIMATE FOOTPRINT INDEX (1)

Model 1 uses the financial inclusion index as a regressor, while Model 2 utilizes individual financial inclusion indicators. Statistical significance at the 1%, 5%, and 10% level is indicated by ***, **, and *, respectively

Figures







Fig. 2 Intra-Regional Financial Inclusion Comparison Source: Global Financial Development Database, 2000-2021

Appendix

SUB-SAHARAN AFRICA COUNTRIES				
Eastern Africa	Western Africa			
Burundi	Benin			
Congo	Burkina Faso			
Congo Republic	Cape Verde			
Kenya	Cameroon			
Madagascar	Chad			
Mauritius	Cote d'Ivoire			
Mozambique	Gabon			
Rwanda	The Gambia			
Sudan	Ghana			
Tanzania	Guinea			
Uganda	Mali			
	Mauritania			
Southern Africa	Niger			
Angola	Nigeria			
Botswana	Senegal			
Eswatini	Sierra Leone			
Lesotho	Тодо			
Malawi				
Namibia				
Seychelles				
South Africa				
Zambia				
Zimbabwe				

Table A1: List of the sample of 38 Sub-Saharan (SSA) economies

Variable	Measurement	Data Source
Environmental Pollution Index	Principal Component Analysis of four (4) variables: CO2 emission, PM 2.5, Methane emission, Nitrous oxide emission	See Appendix Table A3
Financial Inclusion Index	Principal Component Analysis of six (6) variable estimates measuring access, usage, and quality of product	See Appendix Table A4
Energy Use	Per capita electricity generation from oil (kWh)	Our World in Data
Human Development	School enrollment, Tertiary	World Development Indicators 2000 - 2021
Trade Openness	Sum of exports and imports of goods and services as a share of gross domestic product.	World Development Indicators 2000 - 2021
Natural Resource Dependency	Total national resource rents (% of GDP)	World Development Indicators 2000 - 2021
Economic Growth	Gross Domestic Product (GDP) (% growth)	World Development Indicators 2000 - 2021
Government Expenditure	Annual percentage growth of general government final consumption expenditure (% of GDP)	World Development Indicators 2000 - 2021
Urbanisation	Urban Population (% of total population)	World Development Indicators 2000 - 2021

 Table A2: Description of variables and measurement

Table A3: Component variables of the Environmental Pollution Index

Components	Individual indicator(s)	Source of Data
CO2 emissions	CO2 emission per capita (metric tons)	World Development Indicators 2000 – 2021
Particulate Matter (PM) 2.5	PM 2.5 (particles with a diameter of 2.5 micrometres or less)	World Development Indicators 2000 – 2021
Methane emissions	Methane emission (metric tons of CO2 equivalent per capita)	World Development Indicators 2000 – 2021
Nitrous oxide emission	Nitrous oxide emission (metric tons of CO2 equivalent per capita)	World Development Indicators 2000 – 2021

Components	Individual indicator(s)	Source of Data
ATM per 100,000 adults	Number of ATMs per 100,000 adults.	Global Financial Development, 2000-2021
Bank Branches per 100,000 adults	Number of commercial bank branches per 100,000 adults.	Global Financial Development, 2000-2021
Bank Deposits	Money deposited in banks as a share of GDP.	Global Financial Development, 2000-2021
Bank Credit to Bank Deposit	Credit provided by domestic money banks as a share of total deposits.	Global Financial Development, 2000-2021
Life Insurance Premium	Ratio of life insurance premium volume to GDP.	Global Financial Development, 2000-2021
Non-Life Insurance Premium	Ratio of non-life insurance premium volume to GDP	Global Financial Development, 2000-2021

Table A4: Component variables of the Financial Inclusion Index

 Table A5 - Estimation results of the principal component analysis

Eigenvalue of the correlation matrix			Eigenvectors of the first	component	
Component	Eigenvalue	Eigenvalue	Cumulative	Variables	Eigenvector
no.	_	Difference	percentage of		_
			total variance		
(a) Environ	mental Polluti	ion Index			
1	1.649	0.310	0.412	CO2 emission	0.237
2	1.338	0.692	0.747	Particulate Matter (PM) 2.5	0.301
3	0.646	0.280	0.909	Methane emissions	0.674
4	0.366		1.000	Nitrous oxide emission	0.632
(b) Financia	al Inclusion In	dex			
1	3.956	3.000	0.659	ATMs	0.450
2	0.953	0.347	0.818	BBA	0.449
3	0.606	0.292	0.919	Bank credit to Bank deposit	0.221
4	0.314	0.193	0.971	Bank Deposit	0.365
5	0.120	0.068	0.991	Life Insurance	0.451
6	0.052		1.000	Non-life Insurance	0.458

Variable	Observation	Mean	Standard	Minimum	Maximum
			Deviation		
Environmental Pollution Index (EPI)	380	0	1.284	-1.438	7.567
Renewable Energy	608	48.575	36.290	0	100
Financial Inclusion Index (FII)	427	0	1.989	-2.083	9.120
Energy Use	814	193.607	624.345	0	5352.816
Human Development	537	8.637	7.691	0.318	43.959
Trade Openness	805	71.161	35.800	0.757	225.023
Natural Resource	836	10.761	10.435	0.002	59.684
GDP Growth	836	4.012	4.573	-20.599	33.629
Government Expenditure (GE)	769	15.140	6.907	0.952	43.484
Inflation	798	10.431	36.763	-16.860	557.202
Urbanisation	836	39.564	16.562	8.246	90.423

Table A6: Descriptive statistics

	Environmental Pollution Index	Renewable Energy	Financial Inclusion	Energy Use	Human Development	Trade	Natural Resource	Economic Growth	Gov't Expenditure	Urbanization
Environmental Pollution Index	1.000									
Renewable Energy	-0.165	1.000								
Financial Inclusion	0.129	-0.301	1.000							
Energy Use	0.054	-0.147	0.201	1.000						
Human Development	0.162	-0.193	0.534	0.218	1.000					
Trade	0.006	-0.092	0.315	-0.244	0.250	1.000				
Natural Resources	-0.065	0.196	-0.264	-0.223	-0.279	-0.007	1.000			
Economic Growth	0.070	0.034	-0.160	-0.125	-0.208	0.086	0.153	1.000		
Gov't Expenditure	0.081	0.143	0.314	-0.175	0.163	0.503	-0.212	-0.095	1.000	
Urbanization	0.265	-0.280	0.467	0.238	0.512	0.399	0.032	-0.167	0.140	1.000

Table A7: Pairwise	correlations	analysis
--------------------	--------------	----------

*significance at 10% level; **significance at 5% level; ***significance at 1% level

	LINEAR	NON-LINEAR (FII)	INTRA- REGIONAL
Lag of Environmental Pollution Index	0.949***	0.766***	0.694***
	(0.000)	(0.000)	(0.000)
Financial Inclusion	0.016*	0.409***	0.117***
	(0.062)	(0.000)	(0.001)
Financial Inclusion squared	(0000_)	-0.048***	(0.00-)
1		(0.000)	
Financial Inclusion in Southern Africa		(*****)	-1.022***
			(0.000)
Financial Inclusion in Western Africa			-0.992***
			(0.000)
Controls			
Energy Use	0.024	-0.052**	-0.046*
	(0.118)	(0.004)	(0.055)
Human Development	0.003*	-0.022***	-0.016**
	(0.098)	(0.000)	(0.004)
Natural Resources	0.001	0.014***	-0.001
	(0.307)	(0.000)	(0.910)
Trade Openness	-0.001**	-0.012***	-0.001
-	(0.053)	(0.000)	(0.420)
Economic Growth	0.006**	0.003	0.024**
	(0.002)	(0.192)	(0.003)
Government Expenditure	-0.024***	0.031***	-0.003
-	(0.000)	(0.000)	(0.823)
Urbanisation	0.003**	0.019***	-0.002
	(0.026)	(0.000)	(0.568)
Constant	0.284***	-0.111	1.039***
	(0.000)	(0.487)	(0.000)
Diagnostics			
Wald Test	10498.07	93763.17	1988.30
Prob. (Wald)	0.000	0.000	0.000
AR1 (p-value)	0.229	0.221	0.217
AR2 (p-value)	0.369	0.343	0.365
Hansen-J (p-value)	0.613	0.505	0.377
Sargan (p-value)	0.310	0.220	0.216
DHT for instrument			
c. Instruments in levels			
H excluding group	0.256	0.369	
d. IV (years, eq(diff))			
H excluding group	0 573	0.482	0.440
Number of groups	22	0.402	22
Number of instrumental	30	30	30
	162	162	162
	104	102	1.02

Table A8: Effect of Financial Inclusion on Environmental Pollution - based on alternative sets of instruments

IES Working Paper Series

2024

- *1.* Nino Buliskeria, Jaromir Baxa, Tomáš Šestořád: *Uncertain Trends in Economic Policy Uncertainty*
- 2. Martina Lušková: *The Effect of Face Masks on Covid Transmission: A Meta-Analysis*
- *3.* Jaromir Baxa, Tomáš Šestořád: *How Different are the Alternative Economic Policy Uncertainty Indices? The Case of European Countries.*
- *4.* Sophie Ghvanidze, Soo K. Kang, Milan Ščasný, Jon Henrich Hanf: *Profiling Cannabis Consumption Motivation and Situations as Casual Leisure*
- 5. Lorena Skufi, Meri Papavangjeli, Adam Gersl: *Migration, Remittances, and Wage-Inflation Spillovers: The Case of Albania*
- *6.* Katarina Gomoryova: *Female Leadership and Financial Performance: A Meta-Analysis*
- 7. Fisnik Bajrami: *Macroprudential Policies and Dollarisation: Implications for the Financial System and a Cross-Exchange Rate Regime Analysis*
- 8. Josef Simpart: Military Expenditure and Economic Growth: A Meta-Analysis
- 9. Anna Alberini, Milan Ščasný: *Climate Change, Large Risks, Small Risks, and the Value per Statistical Life*
- 10. Josef Bajzík: *Does Shareholder Activism Have a Long-Lasting Impact on Company Value? A Meta-Analysis*
- 11. Martin Gregor, Beatrice Michaeli: *Board Bias, Information, and Investment Efficiency*
- *12.* Martin Gregor, Beatrice Michaeli: *Board Compensation and Investment Efficiency*
- *13.* Lenka Šlegerová: *The Accessibility of Primary Care and Paediatric Hospitalisations for Ambulatory Care Sensitive Conditions in Czechia*
- 14. Kseniya Bortnikova, Tomas Havranek, Zuzana Irsova: *Beauty and Professional Success: A Meta-Analysis*
- 15. Fan Yang, Tomas Havranek, Zuzana Irsova, Jiri Novak: *Where Have All the Alphas Gone? A Meta-Analysis of Hedge Fund Performance*
- 16. Martina Lušková, Kseniya Bortnikova: *Cost-Effectiveness of Women's Vaccination Against HPV: Results for the Czech Republic*
- 17. Tersoo David Iorngurum: Interest Rate Pass-Through Asymmetry: A Meta-Analytical Approach
- 18. Inaki Veruete Villegas, Milan Ščasný: Input-Output Modeling Amidst Crisis: Tracing Natural Gas Pathways in the Czech Republic During the War-Induced Energy Turmoil
- 19. Theodor Petřík: *Distribution Strategy Planning: A Comprehensive Probabilistic* Approach for Unpredictable Environment
- 20. Meri Papavangjeli, Adam Geršl: *Monetary Policy, Macro-Financial Vulnerabilities, and Macroeconomic Outcomes*

- 21. Attila Sarkany, Lukáš Janásek, Jozef Baruník: *Quantile Preferences in Portfolio Choice: A Q-DRL Approach to Dynamic Diversification*
- 22. Jiri Kukacka, Erik Zila: Unraveling Timing Uncertainty of Event-driven Connectedness among Oil-Based Energy Commodities
- 23. Samuel Fiifi Eshun, Evžen Kočenda: *Money Talks, Green Walks: Does Financial Inclusion Promote Green Sustainability in Africa?*

All papers can be downloaded at: http://ies.fsv.cuni.cz



Univerzita Karlova v Praze, Fakulta sociálních věd Institut ekonomických studií [UK FSV – IES] Praha 1, Opletalova 26 E-mail : ies@fsv.cuni.cz http://ies.fsv.cuni.cz